Author's response to reviews

**Title:** Comparison of Contrast in Brightness Mode and Strain Ultrasonography of Glial Brain Tumours

**Authors:**

- Tormod Selbekk (tormod.selbekk@sintef.no)
- Reidar Brekken (reidar.brekken@sintef.no)
- Marit Indergaard (maritki@stud.ntnu.no)
- Ole Solheim (ole.solheim@ntnu.no)
- Geirmund Unsgaard (geirmund.unsgard@ntnu.no)

**Version:** 3 **Date:** 11 April 2012

**Author's response to reviews:** see over
Dear members of the editorial board,

We have made a second revision of the manuscript titled “Comparison of Contrast in Strain- and Brightness Mode Ultrasonography of Glial Brain Tumours” by T. Selbekk, R. Brekken, Marit Indergaard, O. Solheim and G. Unsgaard.

In this revision we have responded to the comment from one of the reviewer's on standardization of the 'reading' method. We have addressed this issue in the Discussion section of the revised manuscript, and added 5 new sentences at the top of page 13 in the manuscript and a new reference to a relevant paper. Please see the revised manuscript for details.

We have also verified that the manuscript should be formatted in accordance with the guidelines of BMC Medical Imaging.

We hope that you find the revised manuscript suitable for publication as Research article in BMC Medical Imaging, and we look forward to your response.

Yours sincerely,

Tormod Selbekk
Senior Research Scientist, SINTEF - Medical Technology Department